Author's response to reviews

Title: Do clinical guidelines reduce clinician dependent costs?

Authors:

George Kosimbei (gkosimbei@gmail.com)
Kara Hanson (Kara.Hanson@lshtm.ac.uk)
Mike English (Menglish@nairobi.kemri-wellcome.org)

Version: 3 Date: 27 May 2011

Author's response to reviews: see over
Response to reviewer’s comments

1. Has been addressed, we are dealing with variable costs which are hospital level.

2. Table 2 is obtained from Grimshaw (2004) and a reference is provided. The authors did not conduct any analysis.